Author's response to reviews

Title: Echocardiographic Predictors of Severe Heart Failure Symptoms in Hypertrophic Cardiomyopathy Patients with Sinus Rhythm

Authors:

Fatih Bayrak (dfatihbayrak@yahoo.com)
Gokhan Kahveci (drmarist@yahoo.co.uk)
Muzaffer Degertekin (mdegertekin@yeditepe.edu.tr)
Bulent Mutlu (dr.bmutlu@superonline.com)

Version: 4  Date: 24 February 2008

Author's response to reviews: see over
Dear Editor,

Please find attached our revised manuscript “Echocardiographic Predictors of Severe Heart Failure Symptoms in Hypertrophic Cardiomyopathy Patients with Sinus Rhythm” that we resubmit after making the necessary format changes required for the Trials journal style. You may find the list of changes below.

We obtained permission to acknowledge from all those mentioned in the acknowledgements. None of the authors of this study had a source of funding.

We look forward to hearing from you at your earliest convenience.

Thank you in anticipation.

Yours sincerely,

Fatih Bayrak MD
The list of changes:

1. Please delete the text "Title:"

"Title:" is deleted.

2. Please delete the short title.

The short title is deleted.

3. Please provide the author names in a list (without qualifications). Please then link each author to their affiliations with superscript numbers.

The necessary changes are performed in accordance with editor’s comments.

4. Please delete the keywords.

Keywords are deleted.

5. Acknowledgements. Please place an Acknowledgements section at the end of the manuscript text, directly preceding the Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

Please list the source(s) of funding for the study, for each author, and for the manuscript.
preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

Acknowledgements section is added as commented by the editor.

6. Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Figure cropping is performed in accordance with the Trials journal style.